Author's response to reviews

Title: Differences in the quality of primary medical care for CVD and Diabetes across the NHS: Evidence from the Quality and Outcomes Framework

Authors:

Gary McLean (gml17y@clinmed.gla.ac.uk)
Bruce Guthrie (b.guthrie@chs.dundee.ac.uk)
Matt Sutton (m.sutton@abdn.ac.uk)

Version: 3 Date: 24 May 2007

Author's response to reviews:

Revised Manuscript: MS1117280466134288

Differences in the quality of primary medical care for CVD and Diabetes across the NHS: Evidence from the Quality and Outcomes Framework

Gary McLean, Matt Sutton, Bruce Guthrie

Dear Editor,

We thank you and the reviewers for their comments. The manuscript has been revised in light of the comments made by reviewer 3. We have changed the order of the tables so they referred to in the text in numerical order (table 7 in previous manuscript is now table 2). We have attached QOF designation to table listing definitions of indicators used and added subheadings of simple process and complex process. The formatting changes requested have also been made.